Author's response to reviews

Title: Influence of calcification on the mechanical stability of plaque in three-dimensional carotid bifurcation model

Authors:

Kelvin KL Wong (k.wong@rmit.edu.au)
Pongpat Thavornpattanapong (Pongpat.Thavornpattanapong@rmit.edu.au)
Sherman CP Cheung (chipok.cheung@rmit.edu.au)
Zhonghua Sun (z.sun@curtin.edu.au)
Jiyuan Tu (Jiyuan.Tu@rmit.edu.au)

Version: 2 Date: 8 November 2011

Author's response to reviews: see over
Dear Sir/Madam,

Submission of “Influence of calcification on the mechanical stability of plaque in three-dimensional carotid bifurcation model”

Please kindly find enclosed the manuscript titled “Influence of calcification on the mechanical stability of plaque in three-dimensional carotid bifurcation model”, by Wong et al. for consideration for publication in BMC Cardiovascular Disorders.

The research work in this paper is original and has not been published anywhere. The authors agree to transfer the right of this work to the publisher of the Journal if the manuscript is accepted for publication. Please address any further correspondences to:

Professor Jiyuan Tu  
School of Aerospace, Mechanical & Manufacturing Engineering  
RMIT University  
PO Box 71 Bundoora VIC 3083  
Tel: 61-3-9925 6191  
Fax: 61-3-9925 6108  
E-mail: jiyuan.tu@rmit.edu.au

Thank you very much for your kind consideration. I look forward to hearing from you.

Yours sincerely,

Jiyuan Tu (PhD)